Author's response to reviews

Title: Intra-muscular myxoid lipoma in the proximal forearm presenting as an olecranon mass with superficial radial nerve palsy: A case report

Authors:

Peter Lewkonia (peter.lewkonia@ucalgary.ca)
Shaun AC Medlicott (shaun.medlicott@cls.ab.ca)
Kevin A Hildebrand (hildebrk@ucalgary.ca)

Version: 4 Date: 25 April 2011

Author's response to reviews: see over
Authors’ comments on revisions:

Thank you for the further suggestions.

The patient’s ethnicity has been added to both the abstract and main case presentation.

The section re-ordering has been completed as instructed in the editor’s email.